Author's response to reviews

Title: Low Income, Community Poverty and Risk of End Stage Renal Disease

Authors:

Deidra C. Crews (dcrews1@jhmi.edu)
Orlando M. Gutiérrez (ogutierr@uab.edu)
Stacey A. Fedewa (sfedewa@emory.edu)
Jean-Christophe Luthi (Jean-Christophe.Luthi@chuv.ch)
David Shoham (dshoham@luc.edu)
Suzanne Judd (sejudd@uab.edu)
Neil R. Powe (npowe@medsfgh.ucsf.edu)
William M. McClellan (wmcclel@emory.edu)

Version: 2  Date: 15 August 2014

Author's response to reviews:

Please see the attached document.